\hypertarget{classUnitToAssign}{\section{\-Unit\-To\-Assign \-Class \-Reference}
\label{classUnitToAssign}\index{\-Unit\-To\-Assign@{\-Unit\-To\-Assign}}
}


{\ttfamily \#include $<$\-Game\-Commander.\-h$>$}



\-Collaboration diagram for \-Unit\-To\-Assign\-:
\nopagebreak
\begin{figure}[H]
\begin{center}
\leavevmode
\includegraphics[width=176pt]{classUnitToAssign__coll__graph}
\end{center}
\end{figure}
\subsection*{\-Public \-Member \-Functions}
\begin{DoxyCompactItemize}
\item 
\hyperlink{classUnitToAssign_a97d7fcdbd1161aa5957cd134fe5c4ab3}{\-Unit\-To\-Assign} (\hyperlink{classBWAPI_1_1Unit}{\-B\-W\-A\-P\-I\-::\-Unit} $\ast$u)
\end{DoxyCompactItemize}
\subsection*{\-Public \-Attributes}
\begin{DoxyCompactItemize}
\item 
\hyperlink{classBWAPI_1_1Unit}{\-B\-W\-A\-P\-I\-::\-Unit} $\ast$ \hyperlink{classUnitToAssign_a9aab2053c9b15ab84603a57a19dbacd5}{unit}
\item 
bool \hyperlink{classUnitToAssign_a4b0e6668fcfb0b20c8be693fbc40e919}{is\-Assigned}
\end{DoxyCompactItemize}


\subsection{\-Constructor \& \-Destructor \-Documentation}
\hypertarget{classUnitToAssign_a97d7fcdbd1161aa5957cd134fe5c4ab3}{\index{\-Unit\-To\-Assign@{\-Unit\-To\-Assign}!\-Unit\-To\-Assign@{\-Unit\-To\-Assign}}
\index{\-Unit\-To\-Assign@{\-Unit\-To\-Assign}!UnitToAssign@{\-Unit\-To\-Assign}}
\subsubsection[{\-Unit\-To\-Assign}]{\setlength{\rightskip}{0pt plus 5cm}{\bf \-Unit\-To\-Assign\-::\-Unit\-To\-Assign} (
\begin{DoxyParamCaption}
\item[{{\bf \-B\-W\-A\-P\-I\-::\-Unit} $\ast$}]{u}
\end{DoxyParamCaption}
)\hspace{0.3cm}{\ttfamily  \mbox{[}inline\mbox{]}}}}\label{classUnitToAssign_a97d7fcdbd1161aa5957cd134fe5c4ab3}


\subsection{\-Member \-Data \-Documentation}
\hypertarget{classUnitToAssign_a4b0e6668fcfb0b20c8be693fbc40e919}{\index{\-Unit\-To\-Assign@{\-Unit\-To\-Assign}!is\-Assigned@{is\-Assigned}}
\index{is\-Assigned@{is\-Assigned}!UnitToAssign@{\-Unit\-To\-Assign}}
\subsubsection[{is\-Assigned}]{\setlength{\rightskip}{0pt plus 5cm}bool {\bf \-Unit\-To\-Assign\-::is\-Assigned}}}\label{classUnitToAssign_a4b0e6668fcfb0b20c8be693fbc40e919}
\hypertarget{classUnitToAssign_a9aab2053c9b15ab84603a57a19dbacd5}{\index{\-Unit\-To\-Assign@{\-Unit\-To\-Assign}!unit@{unit}}
\index{unit@{unit}!UnitToAssign@{\-Unit\-To\-Assign}}
\subsubsection[{unit}]{\setlength{\rightskip}{0pt plus 5cm}{\bf \-B\-W\-A\-P\-I\-::\-Unit}$\ast$ {\bf \-Unit\-To\-Assign\-::unit}}}\label{classUnitToAssign_a9aab2053c9b15ab84603a57a19dbacd5}


\-The documentation for this class was generated from the following file\-:\begin{DoxyCompactItemize}
\item 
\-U\-Alberta\-Bot\-\_\-src/\-Projects/\-U\-Alberta\-Bot/\-Source/\hyperlink{GameCommander_8h}{\-Game\-Commander.\-h}\end{DoxyCompactItemize}
